Author's response to reviews

Title: Induction of osteogenic markers in differentially treated cultures of embryonic stem cells

Authors:

Jorg Handschel (handschel@t-online.de)
Karin Berr (handschel@med.uni-duesseldorf.de)
Rita A Depprich (depprich@med.uni-duesseldorf.de)
Norbert R Kubler (kubler@med.uni-duesseldorf.de)
Christian Naujoks (christian.naujoks@med.uni-duesseldorf.de)
Hans-Peter Wiesmann (HansPeter.Wiesmann@ukmuenster.de)
Michelle Ommerborn (ommerborn@med.uni-duesseldorf.de)
Meyer Ulrich (ulrich.meyer@med.uni-duesseldorf.de)

Version: 2 Date: 2 April 2008

Author's response to reviews:

According to the suggested amendments, especially review 3, we have made the following corrections:

1. We changed the title in: “Induction of osteogenic markers in differentially treated cultures of embryonic stem cells”.

2. We have shortened the introduction.

4. We have briefly focused on the inherent limitations of our study design.